Author’s response to reviews

Title: Myricanol 5-fluorobenzyloxy ether regulation of survivin pathway inhibits human lung adenocarcinoma A549 cells growth in vitro

Authors:

guanhai dai (daiguanhai@gmail.com)
Xuan Chen (chenxuan001564@163.com)
Zeming Ren (rchao007@163.com)
Chenjie Dai (Chenjie.18@intl.zju.edu.cn)
Yeling Tong (tongyeling@sina.com)
Kequn Chai (ckq3301@aliyun.com)

Version: 1 Date: 24 Mar 2020

Author’s response to reviews:

BCAM-D-20-00354R1
Myricanol 5-fluorobenzyloxy ether regulation of survivin pathway inhibits human lung adenocarcinoma A549 cells growth in vitro

Dear editor,

Thank you for your useful comments and suggestions on our manuscript. We have modified the manuscript accordingly, and the detailed corrections are listed below point by point:

1. Please rename Materials and Methods to Methods.
   √ We have renamed Materials and Methods to Methods.

2. Please provide Declaration header.
   √ We have added a Declarations header in the manuscript.

3. Please state clearly the role the funder(s) had in your study in the "funding" section of the declarations.
   √ We have added the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
4. Please include a statement in the Authors' contributions section to the effect that “all authors have read and approved the manuscript”, and ensure that this is the case.

√ We have added “all authors have read and approved the manuscript” in the Authors' contributions section.

Thank you very much for your valuable suggestion to improve our manuscript. The manuscript has been resubmitted to your journal. Our confirm that all author details on the revised version are correct, and all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data. We look forward to your positive response.

Guanhai Dai

Prof of Institute of Basic Medicine
Zhejiang Academy of TCM,
Hangzhou, China